Author’s response to reviews

Title: Population levels of wellbeing and the association with social capital

Authors:
Anne Taylor (anne.taylor@adelaide.edu.au)
Gabrielle Kelly (gabrielle.kelly@sahmri.com)
Eleonora Dal Grande (elenonora.dalgrande@adelaide.edu.au)
David Kelly (david.kelly@sahmri.com)
Tania Marin (tania.marin@sahmri.com)
Nancy Hay (nancy.hay@whatworkswellbeing.org)
Karena Burke (karena.burke@sahmri.com)
Julio Licinio (julio.licinio@sahmri.com)

Version: 5 Date: 13 Jun 2017

Author’s response to reviews:
The required amendments to the manuscript (Individual contributions of all authors and Consent for publication details) have been made.